Author's response to reviews

Title: Superior antigen-specific CD4+ T-cell response with AS03-adjuvantation of a trivalent influenza vaccine in a randomised trial of adults aged 65 and older

Authors:

Robert B Couch (rbcouch@utmb.edu)
Jose M Bayas (jmbayas@clinic.ub.es)
Covadonga Caso (mcaso.hcsc@salud.madrid.org)
Innocent N Mbawuike (mbawuikein@yahoo.com)
Concepcion N Lopez (mnunezl.hulp@salud.madrid.org)
Carine Claey's (carine.x.claeys@gsk.com)
Mohamed El Idrissi (mohamed.s.elidrissi@gsk.com)
CAROLINE Herve (caroline.c.herve@gsk.com)
Beatrice Laupeze (beatrice.n.laupeze@gsk.com)
Lidia Oostvogels (lidia.oostvogels@gsk.com)
Phillipe Moris (philippe.moris@gsk.com)

Version: 5
Date: 11 July 2014

Author's response to reviews: see over
Miss Sheryl Ramos
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom
July 11, 2014

Dear Miss Sheryl Ramos

MS: 2120383938122132
Superior antigen-specific CD4+ T-cell response with AS03-adjuvantation of a trivalent influenza vaccine in a randomised trial of adults aged 65 and older
Robert B Couch, Jose M Bayas, Covadonga Caso, Innocent N Mbauike, Concepcion N Lopez, Carine Claeyis, Mohamed El Idrissi, CAROLINE Herve, Beatrice Laupeze, Lidia Oostvogels and Phillipe Moris

Please find attached a revised CONSORT checklists and amended manuscript to answer the editor’s queries. The NCT number has been corrected to remove the space and the names of the ethical committees are now provided.

We look forward to your acceptance of the manuscript for publication in BMC-ID.

Best regards,

\[Signature\]

Innocent N. Mbauike, Ph.D.
Corresponding Author and Associate Professor,
Department of Molecular Virology and Microbiology,
Baylor College of Medicine,
One Baylor Plaza, BCM280,
Houston, TX 77030, USA.

Telephone: 713-798-6801; Email: mbauike@bcm.tmc.edu